Answer the reviewer #1 of comments as follows:

This expert consensus is very well written. In this consensus relevant domestic and foreign literature on definition, epidemiology, pathogenesis, clinical manifestation, diagnostic criteria, and treatments of myofascial pain syndrome. The consensus is intended to normalize the diagnosis and treatment of myofascial pain syndromeand be used by first-line doctors, including pain physicians to manage patients with myofascial pain syndrome. It will make the clinician better understand of the treatment of the myofascial pain syndrome. Minor comments: 1. The manuscript requires an editing according to the journal's guidelines. 2. Some language polishing should be revised.

Response: 1. Our manuscript has been edited according to the journal's guidelines.

2. The language polishing has been revised.

Answer the reviewer #2 of comment as follows:

Dear authors, Prepare the manuscript in accordance to guidelines for manuscript preparation, after that I will give my opinion about the manuscript quality.

file:///C:/Users/RU/Downloads/Guidelines_for_Manuscript_Preparation_ and_Submission-Review.pdf In your manuscript the text is not written using: -Book Antiqua font -references are not written correctly -check the instruction about preparation of tables (title of table). Reviewer

Response: We have completed the manuscript preparation according to the guide.